Author’s response to reviews

Title: Exploring strategies to reach individuals of Turkish and Moroccan origin for health checks and lifestyle advice: a mixed-methods study

Authors:

Andrea Bukman (sandra.bukman@wur.nl)
Dorit Teuscher (d.teuscher@maastrichtuniversity.nl)
Jamila Ben Meftah (j.ben_meftah@lumc.nl)
Iris Groenenberg (i.groenenberg@lumc.nl)
Mathilde Crone (m.r.crone@lumc.nl)
Sandra van Dijk (s.van_dijk@lumc.nl)
Marieke Bos (m.bos@hartstichting.nl)
Edith Feskens (edith.feskens@wur.nl)

Version: 3 Date: 24 Jun 2016

Author’s response to reviews:

Dear editorial board of BMC Family Practice,

Thank you for message regarding our manuscript, entitled: ‘Exploring strategies to reach individuals of Turkish and Moroccan origin for health checks and lifestyle advice: a mixed-methods study’. In your message, you describe:

“Your manuscript "Exploring strategies to reach individuals of Turkish and Moroccan origin for health checks and lifestyle advice: a mixed-methods study" (FAMP-D-15-00175R2) has been assessed by our editors. We are delighted, in principle, to accept your manuscript for publication in BMC Family Practice. However, before publication, we ask you to make the following minor but essential revisions:

(1) Please change the title of "Introduction" section to "Background"
(2) Please include a section titled "Declarations" and make sure that you include all the required information in our submission guidelines.”

Hereby we send you our revised manuscript. We changed the title of the ‘Introduction’ section to ‘Background’ (line 66). Moreover, we added a ‘Declarations’ section with the required information (lines 521–562).

We hope that the revised manuscript is now suitable for publication in BMC Family Practice. We look forward to hearing from you.

Kind regards,

On behalf of all authors

A.J. Bukman, MSc.